Author's response to reviews

Title: The economic costs of alcohol consumption in Thailand, 2006

Authors:

Montarat Thavorncharoensap (pymbr@mahidol.ac.th)
Yot Teerawattananon (yot.t@hitap.net)
Jomkwan Yothisamut (jomkwan.y@hitap.net)
Chanida Lertpitakpong (chanida@ihpp.thaigov.net)
Khannika Thitiboonsuwan (kikak@yahoo.com)
Prapag Neramitpitakkul (prapag@ihpp.thaigov.net)
Usa Chaikledkaew (usa.c@hitap.net)

Version: 6 Date: 3 June 2010

Author's response to reviews:

On the title page- typo on the word "alcohol"was corrected.
Abstract - "generality" was changed to "generalizability"
- content error " 45.464.6 million bath / 2,804
Background - page 4 the term generality was changed to generalizability
Methods - police cost--> changed font size for 1 sentence and the phrase "As shown in table 1" was deleted
- page 12 deleted . at the end of sentence entitled "respectively were significantly higher than that of the abstainer,."
Results- page 14 As shown in table 2...... 45,464.6 million baht not 45, 464.4
- page 15 It was estimated that there were 39,460 premature deaths,

Table 8: % of GDP for Canada was changed.